Himmelfarb Health Sciences Library, The George Washington University Health Sciences Research Commons

Microbiology, Immunology, and Tropical Medicine Faculty Publications Microbiology, Immunology, and Tropical Medicine

7-1-2015

Levels of 8-OxodG Predict Hepatobiliary Pathology in Opisthorchis viverrini Endemic Settings in Thailand.

Prasert Saichua Khon Kaen University School of Medicine, Khon Kaen, Thailand

Anna Yakovleva George Washington University

Christine Kamamia George Washington University

Amar R. Jariwala George Washington University

Paul J. Brindley George Washington University

See next page for additional authors

Follow this and additional works at: http://hsrc.himmelfarb.gwu.edu/smhs_microbio_facpubs Part of the <u>Medical Immunology Commons</u>, <u>Medical Microbiology Commons</u>, <u>Parasitic</u> <u>Diseases Commons</u>, and the <u>Parasitology Commons</u>

Recommended Citation

Saichua, P., Yakovleva, A., Kamamia, C., Jariwala, A. R., Sithithaworn, J., et al. (2015). Levels of 8-OxodG Predict Hepatobiliary Pathology in Opisthorchis viverrini Endemic Settings in Thailand. PLoS Neglected Tropical Diseases, 9(7), e0003949. doi:10.1371/ journal.pntd.0003949

This Journal Article is brought to you for free and open access by the Microbiology, Immunology, and Tropical Medicine at Health Sciences Research Commons. It has been accepted for inclusion in Microbiology, Immunology, and Tropical Medicine Faculty Publications by an authorized administrator of Health Sciences Research Commons. For more information, please contact hsrc@gwu.edu.

Authors

Prasert Saichua, Anna Yakovleva, Christine Kamamia, Amar R. Jariwala, Paul J. Brindley, Jeffrey M. Bethony, and + 8 more



Citation: Saichua P, Yakovleva A, Kamamia C, Jariwala AR, Sithithaworn J, Sripa B, et al. (2015) Levels of 8-OxodG Predict Hepatobiliary Pathology in *Opisthorchis viverrini* Endemic Settings in Thailand. PLoS Negl Trop Dis 9(7): e0003949. doi:10.1371/ journal.pntd.0003949

Editor: John Pius Dalton, McGill University, CANADA

Received: April 13, 2015

Accepted: July 3, 2015

Published: July 31, 2015

Copyright: © 2015 Saichua et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The raw data used to generate this manuscript are publicly available to all readers at Dryad Digital Data Repository with the accession number doi:10.5061/dryad.pd6mn.

Funding: This research was partially supported by awards from the Royal Golden Jubilee Research Scholarship Program (PHD/0252/2549), Khon Kaen University Research Grant (2013), Higher Education Research Promotion and National Research University Project of Thailand, Office of the Higher Education Commission, through health cluster (SHeP-GMS), R01CA155297 from the National Cancer Institute, P50AI098639 from the National RESEARCH ARTICLE

Levels of 8-OxodG Predict Hepatobiliary Pathology in *Opisthorchis viverrini* Endemic Settings in Thailand

Prasert Saichua^{1,2°}, Anna Yakovleva^{3°}, Christine Kamamia^{3°}, Amar R. Jariwala³, Jiraporn Sithithaworn⁴, Banchob Sripa^{5,6}, Paul J. Brindley³, Thewarach Laha⁷, Eimorn Mairiang⁸, Chawalit Pairojkul^{5,6}, Narong Khuntikeo^{6,9}, Jason Mulvenna¹⁰, Paiboon Sithithaworn^{6,7‡}, Jeffrey M. Bethony^{3‡}*

Biomedical Science Program, Faculty of Graduate School, Khon Kaen University, Khon Kaen, Thailand,
 Department of Preclinical Science, Faculty of Medicine, Thammasat University, Pathumthani, Thailand,
 Department of Microbiology, Immunology and Tropical Medicine, and Research Center for Neglected
 Diseases of Poverty, School of Medicine & Health Sciences, George Washington University, Washington,
 D.C., United States of America, 4 Department of Clinical Microscopy, Faculty of Associated Medical
 Sciences, Khon Kaen University, Khon Kaen, Thailand, 5 Department of Pathology, Faculty of Medicine,
 Khon Kaen University, Khon Kaen, Thailand, 6 Liver Fluke and Cholangiocarcinoma Research Center,
 Faculty of Medicine, Khon Kaen University, Khon Kaen, Thailand, 7 Department of Parasitology, Faculty of
 Medicine, Khon Kaen University, Khon Kaen, Thailand, 8 Department of Radiology, Faculty of Medicine,
 Khon Kaen University, Khon Kaen, Thailand, 9 Department of Surgery, Faculty of Medicine, Khon Kaen,
 University, Khon Kaen, Thailand, 10 Infections Disease Program, QIMR Berghofer Medical Research
 Institute, Brisbane, Queensland, Australia

• These authors contributed equally to this work.

‡ PSi and JMB also contributed equally to this work.

* jbethony@gwu.edu

Abstract

Opisthorchis viverrini is distinct among helminth infections as it drives a chronic inflammatory response in the intrahepatic bile duct that progresses from advanced periductal fibrosis (APF) to cholangiocarcinoma (CCA). Extensive research shows that oxidative stress (OS) plays a critical role in the transition from chronic O. viverrini infection to CCA. OS also results in the excision of a modified DNA lesion (8-oxodG) into urine, the levels of which can be detected by immunoassay. Herein, we measured concentrations of urine 8-oxodG by immunoassay from the following four groups in the Khon Kaen Cancer Cohort study: (1) O. viverrini negative individuals, (2) O. viverrini positive individuals with no APF as determined by abdominal ultrasound, (3) O. viverrini positive individuals with APF as determined by abdominal ultrasound, and (4) O. viverrini induced cases of CCA. A logistic regression model was used to evaluate the utility of creatinine-adjusted urinary 8-oxodG among these groups, along with demographic, behavioral, and immunological risk factors. Receiver operating characteristic (ROC) curve analysis was used to evaluate the predictive accuracy of urinary 8-oxodG for APF and CCA. Elevated concentrations of 8-oxodG in urine positively associated with APF and CCA in a strongly dose-dependent manner. Urinary 8-oxodG concentrations also accurately predicted whether an individual presented with APF or CCA compared to O. viverrini infected individuals without these pathologies. In conclusion, urinary 8-oxodG is a robust 'candidate' biomarker of the progression of APF and CCA from



Institute of Allergy and Infectious Disease, and the National Health and Medical Research Council of Australia (APP1051627). The contents are solely the responsibility of the authors and do not represent the official views of Royal Golden Jubilee Research scholarship, NIAID, NCI, or the NHMRC of Australia. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

chronic opisthorchiasis, which is indicative of the critical role that OS plays in both of these advanced hepatobiliary pathologies. The findings also confirm our previous observations that severe liver pathology occurs early and asymptomatically in residents of *O. viverrini* endemic regions, where individuals are infected for years (often decades) with this food-borne pathogen. These findings also contribute to an expanding literature on 8-oxodG in an easily accessible bodily fluid (e.g., urine) as a biomarker in the multistage process of inflammation, fibrogenesis, and infection-induced cancer.

Author Summary

Opisthorchis viverrini is a food-borne helminth infection that drives a strong inflammatory response in the bile duct that can result in bile duct fibrosis and bile duct cancer (intrahepatic cholangiocarcinoma). Extensive research shows that oxidative stress (OS) plays a critical role in chronic O. viverrini infection transitioning to cancer in the bile duct. OS also results in a modified DNA lesion, referred to as 8-oxodG, excreted in the urine, where it can be detected by an antibody-based test. We measured the concentrations of 8-oxodG in the urine of *O. viverrini*-infected individuals who had developed bile duct fibrosis or bile duct cancer and compared levels of this metabolite in urine to O. viverrini infected individuals who did not have bile duct fibrosis or cancer in Northeastern Thailand. We determined bile duct fibrosis by ultrasonography and bile duct cancer by immunohistochemistry on resected liver tissue. We then built a statistical model to quantify how well urinary 8-oxodG predicted bile duct fibrosis and bile duct cancer in O. viverrini-infected individuals. We found that individuals with elevated levels of 8-oxodG in urine had a greater probability of developing bile duct fibrosis or bile duct cancer from O. viverrini infection. This association occurred in a strongly dose-dependent manner: in other words, the O. viverrini-infected individuals who had the highest concentration of urinary 8oxodG also had the highest risk of presenting with bile duct fibrosis or bile duct cancer. In summary, measuring levels of 8-oxodG in the urine offers a unique opportunity to develop a candidate biomarker for advanced O. viverrini induced hepatobiliary pathologies such as fibrosis and cancer. The findings also confirm our previous observations that severe liver pathology occurs early and asymptomatically in residents of O. viverrini endemic regions, where individuals are infected for years (often decades) with this food-borne neglected tropical diseases (NTD) pathogen.

Introduction

Over 750 million people (10% of the human population) are at risk of infection with food-borne trematodes, with more than 40 million people currently infected with one of three of these parasites: *Clonorchis sinensis*, *Opisthorchis felineus*, *and Opisthorchis viverrini* [1, 2]. *O. viverrini* is considered the most important of these food-borne trematodes due to its well-documented association with hepatobiliary pathologies that include advanced periductal fibrosis (APF) [3, 4] and intrahepatic cholangiocarcinoma (CCA) [5–10]. In Northeastern Thailand (Isaan), uncooked cyprinoid fish, which is the intermediate host for the parasite, are a staple of the diet, with *O. viverrini* infecting an estimated 10 million people in Isaan alone [8]. While infection with *O. viverrini* can be eliminated by chemotherapy (praziquantel), regional culinary practices

result in rapid re-infection after treatment, often leading to life-long infection with the parasite [5, 7, 11] and the highest incidence of CCA in the world (85 per 100,000) [7].

In our community-based ultrasound studies in Northeastern Thailand [8, 9, 12, 13], we have identified a series of pathologic changes that occur early and asymptomatically in the bile duct in individuals resident in *O. viverrini* endemic areas. As individuals can be infected with *O. viverrini* for years (even decades), we hypothesize that a chronic cycle of tissue damage and repair ensues in the intrahepatic biliary ducts as a result of the constant immunological, mechanical and oxidative damage from the parasite, resulting in a persistent "smoldering and chronic inflammatory milieu" [14]. These processes stimulate the production of desmoplastic stroma (i.e., bile duct fibrosis), which has recently been shown to play a crucial role in promoting malignant transformation to CCA (see [15]). In both the hamster and human models of *O. viverrini* infection, fibrosis in the biliary epithelia routinely precedes CCA [8, 10, 16, 17].

The exact mechanism by which stromal desmoplasia (bile duct fibrosis) transforms to CCA is a topic of intense research [15, 17-19]. There is some consensus that an important component in this transformation is the genomic instability that accompanies both fibrogenesis and carcinogenesis[20, 21]. During these processes, cells are recruited to the site of damage and induce a "respiratory burst" from an increased uptake of oxygen, with the accumulation of reactive oxygen species (ROS) referred to as oxidative stress (OS) (see [22]). Both fibrotic [16] and neoplastic transformation [20, 23] have been linked to increased levels of OS by several mechanisms, including DNA damage, genomic instability, and cellular proliferation, The DNA base modifications caused by OS also result in oxidation of guanine residues to 8-oxo-7,8-dihydro-2'-deoxyguanosine (8-oxodG), which are excised into bodily fluids, such as urine, blood and saliva. Augmented 8-oxodG levels in urine have been used as a biomarker for oxidative DNA damage [20, 21, 24–28] in acute lymphoid leukemia, colorectal cancer, high grade cervical dysplasia, scleroderma fibrosis, liver fibrosis, renal cell carcinoma, lung cancers, and prostate cancer [15, 22, 29, 30].

Elevated levels of 8-oxodG have been reported in the urine of individuals chronically infected with O. viverrini and in the urine of individuals with O. viverrini-induced CCA [31-33]. However, to our knowledge, there have been no studies that measure levels of urine 8-oxodG during the interval from initial infection with O. viverrini to neoplastic transformation to CCA, a progression which occurs over years and proceeds through several well-defined hepatobiliary pathologies [12]. The most precisely detected and well documented of these hepatobiliary pathologies is APF [9]. Our community-based ultrasound studies (i.e., Khon Kaen Cancer Cohort or the KKCC) along the Chi River, Khon Kaen Province, Thailand have shown that APF is prevalent among otherwise apparently healthy residents in O. viverrini endemic areas [5, 8-10, 13]. The objective of the current manuscript was to determine if levels of 8-oxodG in urine increased when O. viverrini infection transitioned to APF [34] and if elevated levels of urine 8-oxodG associated with APF were comparable to the markedly elevated levels of urine 8-oxodG observed in individuals with O. viverrini-induced CCA [17, 28, 35]. As APF is a precursor stage to CCA, a threshold concentration of urinary 8-oxodG could then be used to risk stratify APF individuals into those more likely to develop O. viverrini-induced CCA and, as such, take advantage of recent therapeutic advances used to target bile duct fibrosis and the rich milieu it provides for neoplastic transformation to CCA [36]. More specifically, our objectives were as follows: (a) quantify the presence of urinary 8-oxodG in O. viverrini infected individuals with APF compared to O. viverrini-infected individuals without APF and O. viverrini-individuals with CCA and (b) determine the performance of urine 8-oxodG in predicting APF, including its sensitivity, specificity, and ability to predict diagnostic risk (e.g. odds ratios). These studies form the "discovery phase" for a potentially critical and easily accessible candidate biomarker that could be advanced to biomarker validation in large-scale studies in Thailand and in the

neighboring countries of the Mekong River Basin, where *O. viverrini*-induced CCA has the highest incidence in the world [7, 12].

Materials and Methods

Ethics statement

The participants provided written informed consent using forms approved by the Ethics Committee of Khon Kaen University School of Medicine, Khon Kaen, Thailand (reference number HE480528) and the Institutional Review Board (IRB) of the George Washington University School of Medicine, Washington, D.C (GWUMC IRB# 020864). The urine donated by participants from Group 4 was obtained from the biological specimen repository of the Liver Fluke and Cholangiocarcinoma Research Center, Khon Kaen University, Thailand using a protocol approved by the Ethical Committee on Human Research, Faculty of Medicine, Khon Kaen University, Khon Kaen, Thailand (reference Nos. HE450525 and HE531061).

Study population and description of inclusion and exclusion criteria

This study uses baseline data from a recently enrolled village in the Khon Kaen Cancer Cohort (KKCC), which samples villages along the Chi River Basin, in Khon Kaen province, Thailand. A detailed description of the KKCC and the methods used to assemble this cohort have been extensively reported [8–10, 13]. A modification to the original KKCC protocol was approved by the Division of Microbiology and Infectious Diseases (DMID) of the US National Institutes of Health, the Ethics Committee of Khon Kaen University School of Medicine, and the GWU IRB, allowing urine collection, which initiated in 2012 (Fig 1). The current sample of 221 individuals represents the most recent enrollment into the KKCC following this modified protocol (Fig 1). Participant inclusion criteria consisted of enrolling all males and females between 20 and 60 years of age (inclusive) registered with the village health outpost who were willing to



Fig 1. A schematic representation of study participants from Khon Kaen Cancer Cohort (KKCC) and the Liver Fluke and Cholangiocarcinoma Research Center. The KKCC is identified along with approved modifications to the protocol. The exclusion criteria used at various stages are also presented. The stratification of participants into the groups used in the analyses of levels of urine 8-oxodG are depicted.

doi:10.1371/journal.pntd.0003949.g001

participate in the study as evidenced by signing the informed consent form. Participants were excluded from the study if they attended school or worked full-time outside of the village (n = 0) or had a positive urine β -hCG pregnancy test in the case of females (n = 0). Individuals infected with *O. viverrini* were referred to the local public health clinic for treatment with praziquantel (PZQ) regardless of their participation in the study.

The data for the 181 KKCC participants were stratified during analyses to remove thirty two (n = 32) individuals whose feces contained eggs or larvae from other helminths endemic to the region, including hookworm (*Necator americanus*), *Ascaris lumbricoides*, or *Strongyloides stercoralis* (Fig 1). This reduced the study sample to 149 individuals assigned to three groups as follows: Group 1 (n = 23) individuals negative (uninfected) upon fecal examination for *O. viverrini* (also referred to as Endemic Controls or EN); Group 2 (n = 48) *O. viverrini* positive (OV+) individuals who were negative for APF as determined by US (also referred to as Clinical Controls); and Group 3 (n = 78) *O. viverrini* positive individuals who were also positive for APF as determined by US (also referred to as Clinical Cases) (Fig 1). Group 4 consists of samples from 33 individuals who were not part of the KKCC, but had histologically proven *O. viverrini*-associated CCA, whose urine samples were stored at the biological specimen repository of the Liver Fluke and Cholangiocarcinoma Research Center, Khon Kaen University, Thailand. The samples from Group 4 were chosen from the biological specimen repository by matching them on age, sex, and nearest neighbor (i.e., residence at time of death in a village within 10 kilometers of the current study sample) to the Clinical Cases in the KKCC.

Abdominal ultrasound for the detection of advanced periductal fibrosis (APF)

A detailed description of the ultrasonography methods used in this study can be found in the following references [8–10]. Briefly, a mobile, high-resolution ultrasound (US) machine (GE model LOGIQ Book XP) was used. Hepatobiliary abnormalities including portal vein radical echoes, echoes in liver parenchyma, indistinct gallbladder wall, gallbladder size, sludge and suspected CCA were graded and recorded. Individuals were classified as "Non-Advanced Periductal Fibrosis" (APF-) or "controls" if the US grade was 0 or 1, and "Advanced Periductal Fibrosis" (APF+) or "case" if the US grade was 2 or 3. Individuals with alcoholic liver disease, which is seen as fatty liver by US exam, were excluded from this study. Also, individuals with marked hepatic fibrosis not related to OV infection (e.g., cirrhosis from HBV or HCV) were also excluded from this study (see Fig 1).

Collection of samples and their clinical assessment

Collection of feces and urine from participants. Samples of feces and of urine were provided on the same day by each participant in Groups 1–3. Fecal samples were not available for participants in Group 4 (CCA cases) but *O. viverrini* status was noted in their clinical records at the time of death. Diagnosis of infection with *O. viverrini* was established and quantified as eggs per gram of feces (epg) by microscopic examination of feces using the formalin-ethyl acetate concentration technique (FECT), as described in [<u>37</u>] with duplicate examinations of fecal samples.

First morning, mid-stream, urine samples were collected into sterile containers. Unprocessed urine was screened for urinary tract infection by urine strip (ARKRAY'S AUTION Sticks, Japan) and analyzed by an Automated Urine Chemistry Analyzer (AUTION MAX AX-4280, Arkray, USA). Specimens which showed abnormal values were excluded from the study i.e. leukocyte or nitrate positive. Urine samples were centrifuged and the supernatant aliquoted and stored at -20° C. Urinary creatinine concentrations were determined by the Jaffe method [38].

For Group 4 (cases of CCA), urine was age, sex, and nearest-neighbor-matched (by village) using specimens in the repository of the Liver fluke and Cholangiocarcinoma Research Center, Khon Kaen University, Thailand. Participants had been asked to collect first morning urine into clean polypropylene containers, which was held on ice until processed by centrifugation in the laboratory, aliquoted and stored at -20° C.

Collection of serum from study participants. Thirty milliliters of venous blood were collected from individuals in Groups 1, 2 and 3, into siliconized tubes after overnight fasting. Venous blood samples were allowed to clot at room temperature for 30 minutes after collection, centrifuged, and the serum recovered, aliquoted for storage at -20° C in a temperature-monitored freezer. Serum was not available for participants in Group 4 (CCA cases).

Measurement of 8-oxodG in urine. Concentration of 8-oxodG in urine was measured using an HT Trevigen (Gaithersburg, MD) validated competitive ELISA kit. This assay quantifies 8-oxodG in urine using the anti-8-oxodG monoclonal antibody clone 2E2. Briefly 150 µl of 8-OHgG (8-hydroxy-2'-deoxyguanosine) standard were added in triplicate starting at a concentration of 200 nM (56.7 ng/ml) and serially diluted using 1X in phosphate buffered saline (pH 7.2) with 0.1% Tween 20 (PST20) to 3.3 nM (0.89 ng/mL) onto a 96 well microtiter plates pre-coated with 8-oxodG, followed by duplicates of experimental urine samples diluted at 1:20 in PST20. Blank wells with 150 µl of PBST20 and 8-OHgG alone were added in triplicate to each plate as assay controls. Subsequently, 25 µl of monoclonal mouse IgG2b anti-8-oxodG (clone 2E2) conjugated to Keyhole Limpet Hemacyanin (KLH) and horseradish peroxidase (HRP) were diluted in PBST20 and added to each well. The monoclonal mouse IgG2a anti-8-oxodG (clone 2E2), conjugated to KLH specifically binds to 8-hydroxy-2'- deoxyguanosine within DNA as described [39]. The plate was covered and incubated for 1 hour (HR) at room temperature (RT) in the dark. The plates were then washed 6 times using 300 μ /well of 1X PBST20 (7.2 pH) and 100 µl of 3,3',5,5'-Tetramethylbenzidine (TMB) substrate was added to each well and incubated at RT for 15 min in the dark. The reaction was stopped by adding 50 µl 0.2M HCL to the well. Optimal density at 450 nanometers (nm) was determined on a SpectraMax 340PC384 system (Molecular Devices, Sunnyvale CA) using SoftMax Pro Software (Molecular Devices, Sunnyvale CA) for data management. The standard curve was estimated using a polynomial function and level of 8-oxodG in urine were interpolated onto this standard curve and expressed in ng/ml.

Standardization of 8-oxodG by creatinine concentration in urine as determined by the Jaffe method. Measures of creatinine in the urine samples of study participants were determined in ng/mL by the Jaffe method as described above [<u>38</u>]. To standardize the levels of 8-oxodG across participants, level of 8-oxodG was divided by urine creatinine to obtain an adjusted measure of 8-oxodG (in ng/mg of creatinine) [<u>40</u>]. Levels of creatinine were unavailable for 37 individuals in the study. When creatinine levels were unavailable (e.g. for urine from CAA patients), the levels of 8-oxodG were estimated as follows: individuals were separated by sex and infectivity status and their average creatinine concentrations estimated. The creatinine averages were used to adjust the 8-oxodG levels as described above.

Description of the enzyme-linked immunosorbent assay (ELISA) protocol used to quantify levels of the antibody against *O. viverrini* antigen in serum. The ELISA protocol for serum IgG antibody measurements has been previously described in detail [13]. Briefly, flatbottom 96 wells microtiter plate (NUNC, DN) were coated overnight at 4°C with 1 µg/ml of crude somatic *O. viverrini* antigen in a PBS buffer. After washing with a buffer that contained 0.05% Tween 20 in PBS (PBST20), a blocking buffer containing 3 w/v BSA (Fitzgerald, MA) in PBS and 0.5% Tween 20 (Fisher, NJ) was added and incubated at RT for 1 HR. The diluted serum samples were added in duplicate at 100 μ L/well, covered, and incubated overnight at 4°C. After washing with PBST20, 100 μ L of a horseradish peroxidase conjugated goat anti-human IgG (Zymed, CA) diluted in PBST20 was added at 1:1000 and incubated for 2 HR at RT. After incubation and washing with PBST20, the substrate solution which consisted of Orthophenylenediamine (Sigma, MO), 53 mM citric acid anhydrous (Fisher, NJ), 102 mM diasic sodium phosphate dodecahydrate (Acros Organics, NJ) and 30% w/w of hydrogen peroxide (Fisher, NJ) in DI water was added to the plates and incubated at RT in the dark for 30 min. The reaction was stopped by 2N sulfuric acid (BDH, PA) and plates were read using a Spectra-Max 340PC³⁸⁴ microplate reader (Molecular Devices Sunnydale, CA) at 492_{nm}. A standard calibration curve (SCC) made from a pool of high titer human sera against the same crude somatic O. viverrini adult antigen extract was run on each plate in doubling dilutions starting at 1:200 to form a sigmoidal curve when modeled by a 4-parameter logistic (4-PL) using Softmax Pro software (Molecular Devices, Sunnydale, CA). Homologous interpolation between the experimental serum samples and SCC were then used to determine the levels of IgG against the OV crude antigen extract expressed in Arbitrary Units as discussed in detail in Sachiua et al. [13].

Statistical analysis

Data analysis was performed with SAS 9.2 (SAS Institute Inc., Cary, NC, USA). The raw data used to generate this manuscript are publicly available from the Dryad Digital Repository with the accession number doi:10.5061/dryad.pd6mn.

Characteristics of study participants. Descriptive statistics, including number of observations (n), mean, median, standard deviation (SD), and 95% confidence intervals (95% CI) were calculated for demographic variables and baseline characteristics of study participants (<u>Table 1</u>). In instance were a group had less than 5 observations only the median was obtain and reported. The Kruskal-Wallis one-way analysis of variance was used to test for differences in levels of urinary 8-oxodG in study participants stratified by the groups described above (EN, OV+APF-, OV+APF+, or CCA) [<u>41</u>] and to obtain p values. In instances where the test produced a significant result, Dunn's multiple comparisons procedure was used to perform pairwise comparisons of distributions of urinary 8-oxodG levels between the study groups (familywise error rate was controlled at 5%).

Description of the strategy used to assemble the urine 8-oxodG model for corroborating diagnosis with O. viverrini associated pathology. A risk model [42] initially containing potentially relevant demographic [11, 37], behavioral [37, 43], and immunological [6, 44–48] markers was constructed to evaluate the utility of the level of urinary 8-oxodG in predicting APF and CCA status. Our previous reports of elevated levels of serum IgG and IgG1 in individuals with medium to heavy O. viverrini infections prompted us to include serum IgG and IgG1 to O. viverrini crude adult antigen extract as covariates in the biomarker model along with levels of 8-oxodG in the initial stages of model assembly [48]. Age, alcohol consumption (reported and analyzed as a dichotomous variable), and smoking status (reported and analyzed as a dichotomous variable) were also considered as possible covariates [11, 43, 45]. The logistic regression procedure was applied in two parts: in the first phase the model was fit to the data to compute the probability (odds) of receiving a diagnosis of either APF or CCA (categorical) [42], followed by models specifically considering the relationships between EN/APF- participants and either APF+ or CCA participants. The initial model included levels of urinary 8-oxodG, serum levels of IgG and IgG1, and the participants' age, and alcohol consumption habits (equation 1 provided in <u>S1 Text</u>—Supplementary Equations and Definitions). Smoking status was not initially modeled, because the information was not available for CCA participants.

	Median	Mean	SD	95%CI	N	P value ⁺
Sex						
Males	160.17	247.73	235.37	197.86-297.60	88	0.86
Females	148.33	216.16	193.95	176.44-255.89	94	
Age strata						
20–29 ^a	171.40				3	0.43
30–39	201.66	248.87	240.38	144.92-352.82	23	
40–49	132.46	218.71	221.70	162.41-275.01	62	
50+	149.97	235.88	208.80	193.12-278.65	94	
Smoking Status ^b						
No	147.55	201.59	184.30	164.04-239.13	95	0.21
Yes	185.93	270.05	245.28	213.62-326.48	75	
Alcohol Consumption ^c						
No	133.43	176.18	147.62	126.69-225.40	37	0.23
Yes	154.83	237.46	224.40	195.25-279.67	111	
Diagnostic Criteria						
EN (OV-)	103.72	116.18	54.62	92.56-139.80	23	<0.0001
OV+/APF-	126.10	143.77	120.61	108.75-178.80	48	
OV+/APF+	236.08	300.41	245.99	244.94-355.87	78	
CCA	179.77	276.19	239.98	191.10-361.28	33	

Table 1. Creatinine adjusted levels of urine 8-oxodG in the study sample.

OV refers to O. viverrini. APF refers to advanced periductal fibrosis. CCA refers to cholangiocarcinoma.

^a Mean, standard deviation (SD), and 95% confidence interval (CI) not reported when less than 5 participants.

^b Smoking status not available for 12 participants.

^c Alcohol consumption not available for 34 participants.

⁺ Kruskal-Wallis ANOVA used to determine significant differences in levels of urinary 8-oxodG among groups and compute the p values. Significant p-values in bold font.

doi:10.1371/journal.pntd.0003949.t001

We applied stepwise selection, with backward elimination of predictors, to the full fivepredictor model. An initial standard selection criterion of $p \le 0.05$ was used for inclusion of a variable in the model [42, 49]. The backward elimination procedure was initially used to eliminate non-significant predictor variables (p > 0.05) from the model that included all three diagnostically distinct disease levels. Subsequent to this initial modeling step, separate logistic regression models were considered to compare the control individuals with either the APF + individuals or the CCA individuals. These models included predictors that were previously identified as significant and were re-fitted with the step-wise selection procedure and backward elimination to identify the predictors that remained significant for these specific diagnostic models. During this second round of model fitting, predictors with p-values below 0.05 were considered significant and were retained in the model. The final parsimonious models included urinary 8-oxodG as the only significant predictive biomarker. The models are described fully in the results section as well as in equation 2 provided in <u>S1 Text</u>—Supplementary Equations and Definitions.

Odds Ratios (OR) were obtained from these risk models; the OR from the risk model for CCA were adjusted for age.

Elucidating diagnostically informative biomarker levels from the urinary 8-oxodG risk models. Receiver operating characteristic (ROC) curves [49] were constructed by plotting the sensitivity against 1-specificity for the entire range of the possible creatinine adjusted urinary 8-oxodG values of the study participants. Since the ROC curves capture the entire range of the

possible levels of urinary 8-oxodG, it was possible to examine a variety of potential cutpoints, or threshold levels, of the biomarker to identify diagnostically meaningful levels, which may prove relevant in a clinical setting. Sensitivity and specificity values were estimated by the logistic regression procedure for all of the possible cutpoints identified from the ROC curve. Cutpoints that produced simultaneously maximized values of sensitivity and specificity were identified as 'optimal' by the model and were subsequently considered informative and clinically relevant. These threshold values were used to cross-classify the observed and predicted responses of study participants to generate probabilities associated with receiving a positive diagnosis of APF or CCA. Sensitivity and specificity values were estimated from the probabilities by the logistic regression procedure and were subsequently used to evaluate the validity of the diagnostic assay.

Assessing the accuracy of the diagnostic assay, which utilizes informative threshold levels of urinary 8-oxodG, for potential use in the clinical setting. The performance of the diagnostic assay was evaluated by computing the positive predictive value (PPV), negative predictive value (NPV), the likelihood ratio of obtaining a positive test result (LR+), and the likelihood ratio of obtaining a negative test result (LR-). These values were calculated for each relevant diagnostic positivity threshold, using a 50% prevalence rate of *O. viverrini* infection, as determined by fecal exam reported in our previous studies [8–10, 13], in adults between 20 and 60 years of age (inclusive).

The formulae used to calculate measures of assay performance (PPV and NPV) when used in an *O. viverrini*-endemic area [50], as well as the definitions of the LR+ and LR- are included in the supplementary text (Equations 3 and 4 in <u>S1 Text</u>—Supplementary Equations and Definitions).

Results

Descriptive statistics for urinary 8-oxodG in the study sample

Table 1 shows the descriptive statistics for urinary 8-oxodG levels by sex, age strata, smoking status, alcohol consumption, and the criteria used to diagnose various stages of *O. viverrini*-induced infection. The levels were highest among persons in the 30–39 years of age group followed by individuals in the 20–29 years of age group, which consisted of only three individuals. Smokers had higher median levels of 8-oxodG at 185.93ng/mg creatinine compared to 147.55ng/mg creatinine in non-smokers. Similarly, individuals who indicated that they consumed alcohol had higher median levels of creatinine adjusted 8-oxodG compared to those who did not, with 154.83ng/mg creatinine and 133.43ng/mg creatinine, respectively. The 8-oxodG levels increased with advancing disease status.

Creatinine adjusted 8-oxodG levels were significantly elevated in individuals with more advanced hepatobiliary disease

The Kruskal-Wallis one-way analysis of variance demonstrated that there were significant differences in the distributions of urinary 8-oxodG levels between participants separated into groups by disease progression (p < .0001). Dunn's multiple comparisons procedure, for which the family-wise error rate was set at 0.05, identified significant (p < 0.05) differences in the distributions of urinary 8-oxodG between control individuals (both EN and APF-) and *O. viverrini*-infected individuals (APF or CCA positive). No significant differences were observed between participants in EN and APF- groups, which, along with co-occurrence of these individuals in the same population, justified combining these participants into one group in subsequent analyses. No significant difference was observed between participants in the APF+ and



Fig 2. Levels of creatinine corrected urinary 8-oxodG are significantly elevated in *O. viverrini* infected individuals with advanced periductal fibrosis (APF) and *O. viverrini* induced intrahepatic cholangiocarcinoma (CCA). The distributions of 8-oxodG levels in urine are shown in Panel A for all four groups and in Panel B for three groups, with the first group in Panel B being a combination of Endemic Normal (*O. viverrini* negative) and the APF-participants into a single hepatobiliary negative control group. The length of the boxes represents the interquartile range (IQR) or the distance between the 25th and 75th percentiles of each urinary 8-oxodG study group. The median value of each urinary 8-oxodG group are represented by the solid horizontal line in each box; the mean value of the group is depicted by the diamond character in the boxes. The bottom whisker represents the value of the lowest observation in the group. The highest observations in the group are represented by the uppermost circles above the group box; the upper whiskers represents the highest observations within 1.5x IQR. ^aSignificant difference exists between this group and the APF+ individuals; this result was obtained using Dunn's procedure for multiple comparisons following the Kruskal-Wallis one way analysis of variance. ^bSignificant difference exists between this group and the CCA individuals; this result was obtained using Dunn's procedure for multiple comparisons following the Kruskal-Wallis one way analysis of variance.

doi:10.1371/journal.pntd.0003949.g002

CCA groups. Lack of statistically significant findings here may be explained by the presence of three individuals in the APF+ group who had extremely high levels of urinary 8-oxodG (4.66, 4.33, and 4.04 times greater than the group's median urinary 8-oxodG levels). However, the groups remained distinct in subsequent analyses due to the presence of the outlier values and also due to the specific pathological differences that exist between APF+ and CCA individuals.

Fig 2 illustrates the distribution of urinary 8-oxodG in individuals participating in this study. Panel A of Fig 2 depicts the distributions of the levels of urinary 8-oxo-dG in each of the four groups of the study participants and Panel B illustrates the distributions following the amalgamation on participants from groups 1 and 2. Superscripts above the group labels on the x-axis indicate significant findings. Following the combination of control participants into one group (HBP negative), the following results were obtained: the concentrations of 8-oxodG in both cases and controls ranged between 14.25ng/mg creatinine and 1100.97ng/mg creatinine. The range was narrower in the control group (14.25 to 476.58 ng/mg creatinine, n = 71) than among cases, especially OV+ and APF+ (28.89 to 1100.96 ng/mg creatinine, n = 78) and OV +/CCA individuals (26.74 to 933.77 ng/mg creatinine, n = 33). The median 8-oxodG levels of persons with APF or CCA were significantly higher (2.12x and 1.62x, respectively) than the median level of the control group as determined by Dunn's multiple comparisons procedure applied to the Kruskal-Wallis one-way analysis of variance, with the family-wise error rate set at 0.05. These data indicated progressively higher levels of 8-oxodG in the urine during advanced hepatobiliary disease during opisthorchiasis.

Elevated levels of creatinine-adjusted urinary 8-oxodG indicate a significantly increased risk of diagnosis with advanced periductal fibrosis and cholangiocarcinoma in a dose-dependent manner

Various candidate biomarkers, including urinary 8-oxodG, serum level of IgG and IgG1 against a crude adult *O. viverrini* antigen extract were modeled as potential relevant predictors of progression to advanced hepatobiliary pathology during chronic *O. viverrini* infection. In the

	AF	PF Status	CCA Status		
Urine 8-oxodG (ng/mg creatinine)	OR	95% CI	aOR*	95% CI*	
1	1.006	1.004-1.010	1.007	1.003–1.012	
50	1.372	1.201-1.609	1.421	1.167–1.804	
100	1.881	1.442-2.589	2.019	1.362-3.255	
200	3.539	2.079-6.705	4.078	1.854–10.593	

Table 2. Progressively higher urinary 8-oxodG levels show an increasing risk of having received a clinical diagnosis of APF+ or CCA status in *Opisthorchis viverrini* infected individuals.

Odds Ratios (OR) and 95% Confidence Intervals (CI) for creatinine adjusted urinary 8-oxodG levels were computed by the logistic regression procedure form the parsimonious model, which included only significant predictors (p < 0.05). APF refers to advanced periductal fibrosis, and CCA to cholangiocarcinoma.

*In the CCA risk model, age remained a significant predictor, therefore the OR and 95% CI that are reported are adjusted for age or aOR.

doi:10.1371/journal.pntd.0003949.t002

initial stages of model development, control individuals, APF+ individuals, and CCA individuals were included in the model to identify the potential relevant predictors of disease progression. This model identified age (p = 0.0065), serum IgG (p = 0.0002), and urinary 8-oxodG (p < 0.0001) as significant and these predictors were subsequently considered in the two models comparing control individuals to either APF+ participants or CCA participants. In this second stage of model development, urinary 8-oxodG was retained as the single significant biological predictor of an individual having ultrasound confirmed APF ($p \le 0.0001$) or histologically confirmed CCA ($p \le 0.0014$)). Age and serum IgG were no longer significant in the model of APF+ participants (p = 0.4802 and p = 0.2890, respectively). In the model of CCA individuals age remained a significant predictor (p < 0.0001), while serum IgG was eliminated as non-significant (p = 0.0587). The final parsimonious models retain the concentration of creatinine-adjusted urinary 8-oxodG as the only predictive biomarker (for diagnosis of both APF and CCA) along with age as a significant covariate in individuals diagnosed with CCA (equation 2 in <u>S1 Text</u>—Supplementary Equations and Definitions).

The results of a logistic regression model that included urinary 8-oxodG as a predictor were considered alongside the actual ability of the clinical (gold standard) methods to identify specific increases in the level of 8-oxodG in urine of participants in the study. Increasing odds ratios (ORs) were associated with progressively higher concentrations of 8-oxodG, indicting the increased likelihood of progressing to APF and CCA for individuals with higher measurable concentrations of creatinine-adjusted 8-oxodG in the urine. The ORs and their 95% confidence intervals (95% CIs) are summarized in Table 2.

Evaluating the performance of urinary 8-oxodG as a biomarker for advanced periductal fibrosis and cholangiocarcinoma

Receiver operating characteristic (ROC) curves constructed from levels of creatinine-adjusted urinary 8-oxodG measured in study participants are presented in Figs $\underline{3}$ and $\underline{4}$. The area under the ROC curve (AUC) describes the probability of correctly identifying a positive individual as a 'case' and a negative individual as a 'non-case'. An AUC of 1 would describe a diagnostic test that would correctly identify all cases and all non-cases 100% of the time. The 45-degree line in the ROC plot marks the "chance diagonal", which corresponds to an ROC curve with an AUC of 0.50 [50]. Figs $\underline{3}$ and $\underline{4}$ are labeled with the AUC values of each ROC curve; the ROC curve generated from a diagnostic model of APF+ individuals has an AUC of 0.74 and the AUC of the ROC generated from a diagnostic model of CCA individuals is 0.88. This means that levels



Fig 3. Levels of urine 8-oxodG detected by immunoassay can identify *O.viverrini* **infected individuals with advanced periductal fibrosis (APF).** The Receiver Operating Characteristic (ROC) curve illustrates the predictive power of urinary 8-oxodG in identifying APF+ individuals. The predictive power of the assay, measured by the area under curve (AUC) is 74%. The curve is labeled with selected actual amounts of 8-oxodG (ng/mg) detected in study participants; labeled points are presented, at random, in a way to maximize legibility. The model used to construct the ROC curve included negative controls (EN and APF-individuals, n = 71) and APF+ individuals (n = 78). The bold arrow points to the concentration of urine 8-oxodG identified as an optimal diagnostic threshold for corroborating APF status.

doi:10.1371/journal.pntd.0003949.g003

of urinary 8-oxodG in urine correctly identifies individuals with *O. viverrini*-induced APF 74% of the time and correctly identifies individuals with *O. viverrini*-induced CCA 88% of the time. The diagnostic positivity thresholds were established as described in the methods section and are presented in <u>Table 3</u> along with the relevant diagnostic validity parameters (sensitivity and specificity), the PPV, NPV, LR+, and LR- for the urinary 8-oxodG assay.

Discussion

O. viverrini-infected individuals with advanced periductal fibrosis (APF) as determined by abdominal ultrasound had markedly elevated levels of urinary 8-oxodG compared to *O. viver-rini*-infected individuals without APF. Moreover, the concentrations of 8-oxodG in the urine of individuals with APF were comparable to the highly elevated levels of this oxidatively modified DNA lesion in individuals with *O. viverrini*-induced CCA [28, 31–33]. These results clearly



Fig 4. Levels of urine 8-oxodG detected by immunoassay can identify *O.viverrini* infected individuals with intrahepatic cholangiocarcinoma (CCA). The Receiver Operating Characteristic (ROC) curve illustrates the predictive power of urine 8-oxodG in identifying *O. viverrini*-induced CCA individuals. The predictive power of the assay measured by the area under curve (AUC) is 88%. The curve is labeled with selected actual amounts of 8-oxodG (ng/mg) detected in study participants; labeled points are presented, at random, in a way to maximize legibility. The model used to construct the ROC curve included negative controls (EN and APF- individuals, n = 71) and CCA individuals (n = 33). The bold arrow points to the concentration of urine 8 oxodG identified as an optimal diagnostic threshold for corroborating CCA status.

doi:10.1371/journal.pntd.0003949.g004

Table 3. Urinary 8-oxodG diagnostic positivity thresholds (DPT) and corresponding diagnostic validity parameters can be implemented in the clinical setting to detect advanced periductal fibrosis (APF) or cholangiocarcinoma (CCA) in individuals resident in *Opisthorchis viverinni* endemic areas.

	8-oxodG DPT (ng/mg creatinine)	Sensitivity	Specificity	PPV	NPV	LR+	LR-
APF	97	0.72	0.62	0.65	0.69	1.89	0.46
CCA	363	0.85	0.70	0.74	0.82	2.87	0.22

The assay validity parameters including diagnostic sensitivity, diagnostic specificity, PPV, NPV, LR+, and LR- were elucidated from the range of values modeled by the ROC curves in Figs <u>3</u> and <u>4</u>. The process by which diagnostic positivity thresholds were identified is described in detail in the methods section of the manuscript. The values reported reflect the 'optimal' threshold levels identified from the logistic regression model and from the ROC curve analyses. The indicated diagnostic threshold concentrations of urinary 8-oxodG correctly identify APF+ individuals 72% of the time and CCA individuals 85% of the time.

doi:10.1371/journal.pntd.0003949.t003

suggest that elevated levels of this metabolite in urine are indicative of hepatobiliary fibrogenesis and tumorogenesis from chronic *O. viverrini* infection. Moreover, levels of 8-oxodG in the urine of *O. viverrini* infected individuals with APF or CCA individuals corroborated the 'gold' standard diagnostics used to detect both of these hepatobiliary pathologies in a dose-dependent manner: e.g., the highest 50 unit increment of 8-oxodG (200 units) indicated an increased risk of diagnosis of APF or CCA by 354% and 408%, respectively, compared to individuals with no detectable levels of urinary 8-oxodG. Furthermore, the risk models used to evaluate the utility of 8-oxodG as a biomarker were also used to identify the diagnostically relevant levels of 8-oxodG in the urine of study participants. The identification of these diagnostic threshold levels of urinary 8-oxodG, if corroborated in additional larger scale validation studies, would have important implications for urine 8-oxodG as diagnostic tool in field settings, with special significance in resource-limited settings, since they establish benchmarks that may be used to identify individuals at-risk of CCA and refer them for further testing (e.g., confirmatory abdominal ultrasound diagnosis) and preventive chemotherapy.

In order to assess the utility of 8-oxodG as a candidate biomarker for O. viverrini-induced APF and CCA, we also constructed a logistic regression risk model that initially included creatinine-adjusted urinary 8-oxodG levels along with significant demographic, behavioral, and immunological covariates associated with chronic O. viverrini infection. In this risk model, creatinine-adjusted urinary 8-oxodG emerged as the only significant predictor of APF, especially at higher concentrations of 8-oxodG. Creatinine adjusted 8-oxodG was also a significant predictor of CCA status when adjusted for age. Moreover, the risk model showed that 50 unit increases in creatinine-adjusted urinary 8-oxodG (in ng/mg) increased its ability to corroborate APF status by 137% and CCA status by 142% (after adjusting for age) and also helped to establish diagnostically relevant threshold levels of this biomarker. Moreover, elevated levels of urine 8-oxodG accurately identified progression to advanced stages of the disease (Figs 3 and 4), with high odd ratios (ORs) of APF and CCA associated with higher levels of this candidate biomarker (Table 2). As APF is precursor stage to CCA [36], a simple, non-invasive assay for 8-oxodG in urine as a biomarker for this O. viverrini-induced pathology would be of profound benefit in Southeast Asia, especially among populations residing in the resource-limited settings of the Mekong Basin region, where the incidence of intrahepatic CCA is the highest in the world [51].

The methods and standards for measuring urinary 8-oxodG have received considerable attention [24, 40], with a consensus on using urine 8-oxodG as a biomarker established by the European Standards Committee on Urinary (DNA) Lesion Analysis (ESCULA) [24]. When properly detected, measured, and analyzed, ESCULA determined urine 8-oxodG to be a robust, accurate, reproducible, and ("remarkably") stable urine biomarker for OS [24]. The production of 8-hyrdroyguanine is almost exclusively elicited by OS, with the main attack site by oxidative radicals at the N7-C8 bond [26, 52, 53]. While there are findings that suggest that diet contributes to urinary levels of thymine glycol and 8-oxo-7,8-dihydro-guanine (8-oxoGua) [40], little or no information has been reported that this applies to urinary levels of oxidatively modified 2'-deoxyribonucleosides (8-oxodG), hence obviating diet as a potential confounder. As with other studies, the present findings showed a relatively strong association between levels of urinary creatinine and levels of urinary 8-oxodG, which (as explained by Barregard et al. [40]) are likely due to differences in body mass index (BMI), since metabolic rate is associated with lean body mass and a higher metabolic rate generates a larger amount of modified 2-deoxyribonucleoside products in urine [40]: e.g., in general, males excrete more 8-oxodG than females per kilogram body weight [40]. Hence, following the recommendations of Barregard et al. [40] for using 8-oxodG in cross-sectional studies, we normalized urinary 8-oxodG concentrations by individual creatinine concentrations as determined by the Jaffe method [38].

Oxidative DNA damage as a key event in *O. viverrini*-induced CCA has been studied extensively in an animal model (see [23] for review), where oxidatively damaged DNA bases formed along the inflamed intrahepatic biliary ducts, at sites adjacent to the parasite, perhaps from persistent wound repair [31, 32]. Immunohistochemical studies in a hamster model of *O. viverrini*-induced CCA showed that inflammatory cells surrounding the parasite in the bile duct, such as mononuclear cells and eosinophils, generate reactive oxygen species (ROS), which induce oxidative stress (OS) and increased cleavage of 8-oxodG [31, 32]. Additionally, immunohistochemical analyses on livers resected from humans with *O. viverrini*-induced CCA showed 8-oxodG in tumor tissue from the bile duct [20, 33]. Moreover, Thana et al. [33] observed elevated levels of 8-oxdG in the urine of individuals infected with *O. viverrini* compared to healthy controls as also observed in the current study (Fig 2 Panel A). However, the current study adds to the literature the observation that urine 8-oxodG can be detect in markedly higher concentration when *O. viverrini* infected individuals who have progressed to APF or CCA (Fig 2).

The observation that urinary 8-oxodG concentrations were similar in APF and CCA individuals support our hypothesis that common mechanisms drive bile duct fibrosis and bile duct tumorogenesis from chronic *O. viverrini* infection [8–10, 12, 13]. It is also in keeping with findings from other groups that chronic bile duct inflammation leads to a desmoplastic stroma (i.e., fibrosis) in the bile duct that precedes CCA (reviewed by Sirica and Gores [36]. As depicted in S1 Fig, elevated levels of 8-oxodG in the urine of APF individuals reflects ongoing tissue repair and the smoldering inflammatory milieu" [14] in the hepatobiliary epithelia from persistent injury from the parasite [15, 30, 54, 55]. This "desmoplastic reaction" to chronic *O. viverrini* infection provides a rich niche for cancer cells to develop and progress [1, 36, 56, 57]. Currently, therapeutic targeting to reduce desmoplastic stroma (periductal fibrosis) to prevent CCA is being investigated [36, 56, 57].

Taken together, these data show that urine 8-oxodG may be an excellent biomarker for the advanced hepatobiliary pathology that occurs prior to O. viverrini-induced CCA. In keeping with a recent position statement from the European Group on Tumor Markers [58], the current manuscript places urine 8-oxodG as a "candidate biomarker" for APF and CCA in the "discovery phase", i.e., "when differential expression of a specific marker is shown to associate with a 'gold' standard clinical outcome". The next phase of biomarker development for urine 8-oxodG is the "verification stage", when our analyses would be extended to a much larger sample (i.e., hundreds) of individuals infected with O. viverrini and at risk of CCA. The objective of the verification stage would be to incorporate the broadest range of cases and controls in order to capture the environmental, genetic, biological, and stochastic variation in the population in the Mekong Basin Subregion. As the diagnostic sensitivity of the candidate biomarker has been established herein, the verification stage would focus on the specificity of the candidate biomarker and the utility of the diagnostic threshold levels set forth in this manuscript [58]. If validated, levels of urinary 8-oxodG may be an inexpensive way to identify at-risk individuals, who may be referred for subsequent, more demanding testing. This process would streamline diagnostics for APF and CCA and perhaps improve the utilization of the limited resources allocated to cancer screening in this region.

Conclusion

The findings herein confirm previous observations that severe hepatobiliary disease occurs early and asymptomatically among residents in *O. viverrini* endemic areas. A simple, non-invasive assay targeting 8-oxodG in urine would be of profound benefit to populations in Southeast Asia, especially in the resource-limited settings of the Mekong Basin region countries of

Thailand, Laos and Cambodia, where the incidence of *O. viverrini*-induced CCA is the highest in the world [7]. The future plan for the candidate biomarker includes moving to a verification step to test its accuracy in a larger sample size [58].

Supporting Information

S1 Fig. 8-oxodG is a biomarker of disease progression in individuals chronically infected with *Opisthorchis viverrini*. An illustration of the effect of chronic inflammatory challenge experienced by OV-infected individuals and its relationship to oxidative stress as determined by levels of 8-oxodG in urine. Increases in the levels of creatinine-adjusted 8-oxodG in urine are biomarkers for the risk of advanced periductal fibrosis (APF) and cholangiocarcinoma (CCA).

(TIF)

S1 Text. Supplementary equations and definitions. This file includes equations describing the logistic regression models used in this manuscript, equations for computing assessment measures of diagnostic accuracy of the urine 8-oxodG diagnostic assay, as well as definitions of relevant epidemiologic terminology used in evaluating assay performance. (DOCX)

Acknowledgments

We thank the people in the *O. viverrini* endemic areas of Khon Kaen City, Khon Kaen province Thailand for their participant in this project.

Author Contributions

Conceived and designed the experiments: PSa AY CK BS TL EM JS PJB JMB. Performed the experiments: PSa BS TL EM PSi JMB. Analyzed the data: PSa AY CK ARJ PSi JMB. Contributed reagents/materials/analysis tools: CP NK. Wrote the paper: PSa AY CK ARJ JS BS PJB JM PSi JMB.

References

- Furst T, Duthaler U, Sripa B, Utzinger J, Keiser J. Trematode infections: liver and lung flukes. Infectious disease clinics of North America. 2012; 26(2):399–419. Epub 2012/05/29. doi: <u>10.1016/j.idc.2012.03</u>.
 <u>008</u> PMID: <u>22632646</u>
- Furst T, Keiser J, Utzinger J. Global burden of human food-borne trematodiasis: a systematic review and meta-analysis. The Lancet Infectious diseases. 2012; 12(3):210–21. Epub 2011/11/24. doi: <u>10.</u> <u>1016/S1473-3099(11)70294-8</u> PMID: <u>22108757</u>
- Mairiang E, Elkins DB, Mairiang P, Chaiyakum J, Chamadol N, Loapaiboon V, et al. Relationship between intensity of *Opisthorchis viverrini* infection and hepatobiliary disease detected by ultrasonography. Journal of gastroenterology and hepatology. 1992; 7(1):17–21. Epub 1992/01/01. PMID: <u>1311966</u>
- Mairiang E, Mairiang P. Clinical manifestation of opisthorchiasis and treatment. Acta tropica. 2003; 88(3):221–7. Epub 2003/11/13. PMID: <u>14611876</u>
- Sripa B. Pathobiology of opisthorchiasis: an update. Acta tropica. 2003; 88(3):209–20. Epub 2003/11/ 13. PMID: <u>14611875</u>
- Pinlaor S, Sripa B, Sithithaworn P, Yongvanit P. Hepatobiliary changes, antibody response, and alteration of liver enzymes in hamsters re-infected with *Opisthorchis viverrini*. Experimental parasitology. 2004; 108(1–2):32–9. Epub 2004/10/20. d PMID: <u>15491546</u>
- Sripa B, Kaewkes S, Sithithaworn P, Mairiang E, Laha T, Smout M, et al. Liver fluke induces cholangiocarcinoma. PLoS medicine. 2007; 4(7):e201. Epub 2007/07/12. PMID: <u>17622191</u>
- 8. Sripa B, Mairiang E, Thinkhamrop B, Laha T, Kaewkes S, Sithithaworn P, et al. Advanced periductal fibrosis from infection with the carcinogenic human liver fluke *Opisthorchis viverrini* correlates with

elevated levels of interleukin-6. Hepatology. 2009; 50(4):1273–81. Epub 2009/08/14. doi: <u>10.1002/hep.</u> <u>23134</u> PMID: <u>19676135</u>

- Mairiang E, Laha T, Bethony JM, Thinkhamrop B, Kaewkes S, Sithithaworn P, et al. Ultrasonography assessment of hepatobiliary abnormalities in 3359 subjects with *Opisthorchis viverrini* infection in endemic areas of Thailand. Parasitology international. 2012; 61(1):208–11. Epub 2011/07/21. doi: <u>10.</u> <u>1016/j.parint.2011.07.009</u> PMID: <u>21771664</u>
- Sripa B, Thinkhamrop B, Mairiang E, Laha T, Kaewkes S, Sithithaworn P, et al. Elevated plasma IL-6 associates with increased risk of advanced fibrosis and cholangiocarcinoma in individuals infected by *Opisthorchis viverrini*. PLoS neglected tropical diseases. 2012; 6(5):e1654. Epub 2012/05/26. doi: <u>10.</u> <u>1371/journal.pntd.0001654</u> PMID: <u>22629477</u>
- Sithithaworn P, Haswell-Elkins M. Epidemiology of *Opisthorchis viverrini*. Acta tropica. 2003; 88(3): 187–94. Epub 2003/11/13. P PMID: <u>14611873</u>
- 12. Sripa B, Brindley PJ, Mulvenna J, Laha T, Smout MJ, Mairiang E, et al. The tumorigenic liver fluke *Opisthorchis viverrini*—multiple pathways to cancer. Trends in parasitology. 2012; 28(10):395–407. Epub 2012/09/06. doi: <u>10.1016/j.pt.2012.07.006</u> PMID: <u>22947297</u>
- Saichua P, Sithithaworn P, Jariwala AR, Diemert DJ, Sithithaworn J, Sripa B, et al. Microproteinuria during *Opisthorchis viverrini* infection: a biomarker for advanced renal and hepatobiliary pathologies from chronic opisthorchiasis. PLoS neglected tropical diseases. 2013; 7(5):e2228. Epub 2013/05/30. doi: 10.1371/journal.pntd.0002228 PMID: 23717698
- Balkwill F, Charles KA, Mantovani A. Smoldering and polarized inflammation in the initiation and promotion of malignant disease. Cancer cell. 2005; 7(3):211–7. Epub 2005/03/16. PMID: <u>15766659</u>
- Pellicoro A, Ramachandran P, Iredale JP, Fallowfield JA. Liver fibrosis and repair: immune regulation of wound healing in a solid organ. Nature reviews Immunology. 2014; 14(3):181–94. Epub 2014/02/26. doi: 10.1038/nri3623 PMID: 24566915
- 16. Sripa B, Bethony JM, Sithithaworn P, Kaewkes S, Mairiang E, Loukas A, et al. Opisthorchiasis and Opisthorchis-associated cholangiocarcinoma in Thailand and Laos. Acta tropica. 2011; 120 Suppl 1: S158–68. Epub 2010/07/27. doi: 10.1016/j.actatropica.2010.07.006 PMID: 20655862
- Thanan R, Oikawa S, Yongvanit P, Hiraku Y, Ma N, Pinlaor S, et al. Inflammation-induced protein carbonylation contributes to poor prognosis for cholangiocarcinoma. Free radical biology & medicine. 2012; 52(8):1465–72. Epub 2012/03/02.
- Dvorak HF. Tumors: wounds that do not heal. Similarities between tumor stroma generation and wound healing. The New England journal of medicine. 1986; 315(26):1650–9. Epub 1986/12/25. PMID: 3537791
- Schafer M, Werner S. Cancer as an overhealing wound: an old hypothesis revisited. Nature reviews Molecular cell biology. 2008; 9(8):628–38. Epub 2008/07/17. doi: 10.1038/nrm2455 PMID: 18628784
- Murata M, Thanan R, Ma N, Kawanishi S. Role of nitrative and oxidative DNA damage in inflammationrelated carcinogenesis. Journal of biomedicine & biotechnology. 2012; 2012:623019. Epub 2012/03/ 01.
- Ohnishi S, Ma N, Thanan R, Pinlaor S, Hammam O, Murata M, et al. DNA damage in inflammationrelated carcinogenesis and cancer stem cells. Oxidative medicine and cellular longevity. 2013; 2013:387014. Epub 2014/01/03. doi: <u>10.1155/2013/387014</u> PMID: <u>24382987</u>
- Gabrielli A, Svegliati S, Moroncini G, Amico D. New insights into the role of oxidative stress in scleroderma fibrosis. The open rheumatology journal. 2012; 6:87–95. Epub 2012/07/18. doi: <u>10.2174/</u> <u>1874312901206010087</u> PMID: <u>22802906</u>
- Yongvanit P, Pinlaor S, Bartsch H. Oxidative and nitrative DNA damage: key events in opisthorchiasisinduced carcinogenesis. Parasitology international. 2012; 61(1):130–5. Epub 2011/06/28. doi: <u>10.</u> <u>1016/j.parint.2011.06.011</u> PMID: <u>21704729</u>
- Evans MD, Olinski R, Loft S, Cooke MS. Toward consensus in the analysis of urinary 8-oxo-7,8dihydro-2'-deoxyguanosine as a noninvasive biomarker of oxidative stress. FASEB journal: official publication of the Federation of American Societies for Experimental Biology. 2010; 24(4):1249–60. Epub 2009/12/08.
- 25. Evans MD, Saparbaev M, Cooke MS. DNA repair and the origins of urinary oxidized 2'-deoxyribonucleosides. Mutagenesis. 2010; 25(5):433–42. Epub 2010/06/05. doi: <u>10.1093/mutage/geq031</u> PMID: <u>20522520</u>
- Lee KF, Chung WY, Benzie IF. Urine 8-oxo-7,8-dihydro-2'-deoxyguanosine (8-oxodG), a specific marker of oxidative stress, using direct, isocratic LC-MS/MS: Method evaluation and application in study of biological variation in healthy adults. Clinica chimica acta; international journal of clinical chemistry. 2010; 411(5–6):416–22. Epub 2009/12/29. doi: 10.1016/j.cca.2009.12.013 PMID: 20035732

- Broedbaek K, Weimann A, Stovgaard ES, Poulsen HE. Urinary 8-oxo-7,8-dihydro-2'-deoxyguanosine as a biomarker in type 2 diabetes. Free radical biology & medicine. 2011; 51(8):1473–9. Epub 2011/08/ 09.
- Thanan R, Pairojkul C, Pinlaor S, Khuntikeo N, Wongkham C, Sripa B, et al. Inflammation-related DNA damage and expression of CD133 and Oct3/4 in cholangiocarcinoma patients with poor prognosis. Free radical biology & medicine. 2013; 65:1464–72. Epub 2013/08/07.
- Bataller R, Brenner DA. Liver fibrosis. The Journal of clinical investigation. 2005; 115(2):209–18. Epub 2005/02/04. PMID: <u>15690074</u>
- Seki E, Schwabe RF. Hepatic Inflammation and Fibrosis: Functional Links and Key Pathways. Hepatology. 2014. Epub 2014/07/30.
- Pinlaor S, Hiraku Y, Ma N, Yongvanit P, Semba R, Oikawa S, et al. Mechanism of NO-mediated oxidative and nitrative DNA damage in hamsters infected with *Opisthorchis viverrini*: a model of inflammation-mediated carcinogenesis. Nitric oxide: biology and chemistry / official journal of the Nitric Oxide Society. 2004; 11(2):175–83. Epub 2004/10/20.
- **32.** Pinlaor S, Ma N, Hiraku Y, Yongvanit P, Semba R, Oikawa S, et al. Repeated infection with *Opisthorchis viverrini* induces accumulation of 8-nitroguanine and 8-oxo-7,8-dihydro-2'-deoxyguanine in the bile duct of hamsters via inducible nitric oxide synthase. Carcinogenesis. 2004; 25(8):1535–42. Epub 2004/04/03. PMID: <u>15059927</u>
- 33. Thanan R, Murata M, Pinlaor S, Sithithaworn P, Khuntikeo N, Tangkanakul W, et al. Urinary 8-oxo-7,8dihydro-2'-deoxyguanosine in patients with parasite infection and effect of antiparasitic drug in relation to cholangiocarcinogenesis. Cancer epidemiology, biomarkers & prevention: a publication of the American Association for Cancer Research, cosponsored by the American Society of Preventive Oncology. 2008; 17(3):518–24. Epub 2008/03/20.
- Chamadol N, Pairojkul C, Khuntikeo N, Laopaiboon V, Loilome W, Sithithaworn P, et al. Histological confirmation of periductal fibrosis from ultrasound diagnosis in cholangiocarcinoma patients. Journal of hepato-biliary-pancreatic sciences. 2014; 21(5):316–22. Epub 2014/01/15. doi: <u>10.1002/jhbp.64</u> PMID: <u>24420706</u>
- 35. Thanan R, Oikawa S, Hiraku Y, Ohnishi S, Ma N, Pinlaor S, et al. Oxidative Stress and Its Significant Roles in Neurodegenerative Diseases and Cancer. International journal of molecular sciences. 2014; 16(1):193–217. Epub 2014/12/31. doi: 10.3390/ijms16010193 PMID: 25547488
- Sirica AE, Gores GJ. Desmoplastic stroma and cholangiocarcinoma: clinical implications and therapeutic targeting. Hepatology. 2014; 59(6):2397–402. Epub 2013/10/15. doi: <u>10.1002/hep.26762</u> PMID: 24123296
- Haswell-Elkins MR, Elkins DB, Sithithaworn P, Treesarawat P, Kaewkes S. Distribution patterns of Opisthorchis viverrini within a human community. Parasitology. 1991; 103 Pt 1:97–101. Epub 1991/08/ 01. PMID: <u>1945529</u>
- Myers GL, Miller WG, Coresh J, Fleming J, Greenberg N, Greene T, et al. Recommendations for improving serum creatinine measurement: a report from the Laboratory Working Group of the National Kidney Disease Education Program. Clinical chemistry. 2006; 52(1):5–18. Epub 2005/12/08. PMID: 16332993
- Soultanakis RP, Melamede RJ, Bespalov IA, Wallace SS, Beckman KB, Ames BN, et al. Fluorescence detection of 8-oxoguanine in nuclear and mitochondrial DNA of cultured cells using a recombinant Fab and confocal scanning laser microscopy. Free radical biology & medicine. 2000; 28(6):987–98. Epub 2000/05/10.
- Barregard L, Moller P, Henriksen T, Mistry V, Koppen G, Rossner P Jr., et al. Human and methodological sources of variability in the measurement of urinary 8-oxo-7,8-dihydro-2'-deoxyguanosine. Antioxidants & redox signaling. 2013; 18(18):2377–91. Epub 2012/12/04.
- 41. Agresti A. Categorical Data Analysis. Hoboken, NJ: Wiley; 2013.
- Glas AS, Lijmer JG, Prins MH, Bonsel GJ, Bossuyt PM. The diagnostic odds ratio: a single indicator of test performance. Journal of clinical epidemiology. 2003; 56(11):1129–35. Epub 2003/11/15. PMID: <u>14615004</u>
- 43. Honjo S, Srivatanakul P, Sriplung H, Kikukawa H, Hanai S, Uchida K, et al. Genetic and environmental determinants of risk for cholangiocarcinoma via *Opisthorchis viverrini* in a densely infested area in Nakhon Phanom, northeast Thailand. International journal of cancer Journal international du cancer. 2005; 117(5):854–60. Epub 2005/06/16. PMID: <u>15957169</u>
- 44. Elkins DB, Sithithaworn P, Haswell-Elkins M, Kaewkes S, Awacharagan P, Wongratanacheewin S. Opisthorchis viverrini: relationships between egg counts, worms recovered and antibody levels within an endemic community in northeast Thailand. Parasitology. 1991; 102 Pt 2:283–8. Epub 1991/04/01. PMID: <u>1852494</u>

- 45. Haswell-Elkins MR, Sithithaworn P, Mairiang E, Elkins DB, Wongratanacheewin S, Kaewkes S, et al. Immune responsiveness and parasite-specific antibody levels in human hepatobiliary disease associated with *Opisthorchis viverrini* infection. Clinical and experimental immunology. 1991; 84(2):213–8. Epub 1991/05/01. PMID: 2025950
- Akai PS, Pungpak S, Chaicumpa W, Viroj K, Bunnag D, Befus AD. Serum antibody response to Opisthorchis viverrini antigen as a marker for opisthorchiasis-associated cholangiocarcinoma. Transactions of the Royal Society of Tropical Medicine and Hygiene. 1994; 88(4):471–4. Epub 1994/07/01. PMID: 7570848
- 47. Tesana S, Srisawangwong T, Sithithaworn P, Itoh M, Phumchaiyothin R. The ELISA-based detection of anti-Opisthorchis viverrini IgG and IgG4 in samples of human urine and serum from an endemic area of north-eastern Thailand. Annals of tropical medicine and parasitology. 2007; 101(7):585–91. Epub 2007/09/20. PMID: 17877877
- 48. Pinlaor P, Pongsamart P, Hongsrichan N, Sangka A, Srilunchang T, Mairiang E, et al. Specific serum IgG, but not IgA, antibody against purified *Opisthorchis viverrini* antigen associated with hepatobiliary disease and cholangiocarcinoma. Parasitology international. 2012; 61(1):212–6. Epub 2011/07/02. doi: 10.1016/j.parint.2011.06.014 PMID: 21718798
- 49. Gönen M. Analyzing Receiver Operating Characteristic Curves with SAS. Cary, NC: SAS Publishers; 2007.
- 50. Zhou X, McClish D, Obuchowski N. Statistical Methods in Diagnostic Medicine. Hoboken, NJ: Wiley; 2011.
- 51. IARC. A review of human carcinogens. Part B: Biological agents Lyon, France: World Health Organization; 2012.
- 52. Cooke MS, Olinski R, Loft S. Measurement and meaning of oxidatively modified DNA lesions in urine. Cancer epidemiology, biomarkers & prevention: a publication of the American Association for Cancer Research, cosponsored by the American Society of Preventive Oncology. 2008; 17(1):3–14. Epub 2008/01/18.
- Loft S, Danielsen P, Lohr M, Jantzen K, Hemmingsen JG, Roursgaard M, et al. Urinary excretion of 8oxo-7,8-dihydroguanine as biomarker of oxidative damage to DNA. Archives of biochemistry and biophysics. 2012; 518(2):142–50. Epub 2012/01/14. doi: 10.1016/j.abb.2011.12.026 PMID: 22239988
- Castello G, Scala S, Palmieri G, Curley SA, Izzo F. HCV-related hepatocellular carcinoma: From chronic inflammation to cancer. Clin Immunol. 2010; 134(3):237–50. Epub 2009/11/17. doi: <u>10.1016/j. clim.2009.10.007</u> PMID: <u>19910258</u>
- 55. Cernaro V, Lacquaniti A, Donato V, Fazio MR, Buemi A, Buemi M. Fibrosis, regeneration and cancer: what is the link? Nephrology, dialysis, transplantation: official publication of the European Dialysis and Transplant Association—European Renal Association. 2012; 27(1):21–7. Epub 2011/11/22.
- Fingas CD, Bronk SF, Werneburg NW, Mott JL, Guicciardi ME, Cazanave SC, et al. Myofibroblastderived PDGF-BB promotes Hedgehog survival signaling in cholangiocarcinoma cells. Hepatology. 2011; 54(6):2076–88. Epub 2011/11/01. doi: 10.1002/hep.24588 PMID: 22038837
- Fingas CD, Mertens JC, Razumilava N, Bronk SF, Sirica AE, Gores GJ. Targeting PDGFR-beta in Cholangiocarcinoma. Liver international: official journal of the International Association for the Study of the Liver. 2012; 32(3):400–9. Epub 2011/12/03.
- Duffy MJ, Sturgeon CM, Soletormos G, Barak V, Molina R, Hayes DF, et al. Validation of New Cancer Biomarkers: A Position Statement from the European Group on Tumor Markers. Clinical chemistry. 2015; 61(6):809–20. Epub 2015/04/18. PMID: <u>25882892</u>